Author's response to reviews

Title: Need for a standardized protocol for stress echocardiography in provoking subaortic gradient in various cardiac conditions

Authors:

Pawel Petkow Dimitrow (dimitrow@mp.pl)
Carlos Cotrim (carlosadcotrim@hotmail.com)
Tsung O Cheng (tcheng@mfa.gwu.edu)

Version: 3
Date: 23 May 2014

Author's response to reviews:

Dear Editor

Please find enclosed corrected version of our manuscript to consider for publication in your Journal. New corrections have been marked by green color. The following changes we propose:

1. And 2. In Table 2 we have described clinical impact of performing exercise stress echocardiography in various conditions

2. = 3. The pathophysiologic reasons responsible for LVOTG in various conditions have been displayed in Schemata 2.

Sincerely

PP Dimitrow